Author's response to reviews

Title: A Novel Single Base Pair Duplication in WDR62 Causes Primary Microcephaly

Authors:

Verena Rupp (mirwasay@live.com)
Sobiah Rauf (s.rauf87@gmail.com)
Windpassinger Christian (mir77uspk@yahoo.com)
Asif Mir (asif.mir@iiu.edu.pk)

Version: 2 Date: 7 May 2014

Author's response to reviews:

Hi

We are submitting our paper as "case study" as per your directives. We have the study consent of affected and normal individuals and editor can see these any time. We have also given the image permission by the study participants

regards

Asif mir.